Author’s response to reviews

Title: Impact of urine and mixed incontinence on long-term care preference: A vignette-survey study of community-dwelling older adults

Authors:
Nicolas Carvalho (nicolas.carvalho@chuv.ch; nic.carvalh@gmail.com)
Sarah Fustinoni (Sarah.Fustinoni@chuv.ch)
Nazanin Abolhassani (Nazanin.Abolhassani@chuv.ch)
Juan Manuel Blanco (Juan-Manuel.Blanco@chuv.ch)
Lionel Meylan (Lionel.Meylan@chuv.ch)
Brigitte Santos-Eggimann (Brigitte.Santos-Eggimann@chuv.ch)

Version: 2 Date: 19 Jan 2020

Author’s response to reviews:

Dear Editor,

Please find enclosed our responses to the lastest revisions.

We would like to thank once again the reviewers and editor for careful and thorough reading of this manuscript and for the thoughtful comments and constructive suggestions, which help to improve the quality of this manuscript.